Author's response to reviews

Title: Psychiatric Disorder Criteria and their Application to Research in Different Racial Groups

Authors:

Nancy C Low (lown@mail.nih.gov)
John Hardy (hardyi@mail.nih.gov)

Version: 4 Date: 22 December 2006

Author's response to reviews:

YOUR REQUEST

1. Manuscript sections - Manuscript sections should include (in the following order): Abstract, Background, Results, Discussion, Conclusions, Methods, Abbreviations, Authors' contributions, Acknowledgements, References, Figure legends (if any), Tables (if any), Description of Additional files (if any).

OUR RESPONSE: THIS ARTICLE TYPE IS A CORRESPONDENCE TYPE ARTICLE, THEREFORE, IT DOESN'T CONTAIN THE SECTION TITLES OF "BACKGROUND, RESULTS, DISCUSSION, CONCLUSIONS, METHODS" IN THE TEXT OF THE MANUSCRIPT. WE EXAMINED BMC PSYCHIATRY FOR OTHER EXAMPLES OF CORRESPONDENCE TYPE ARTICLES AND BELIEVE OUR FORMAT IS THE SAME AS THOSE.

2. Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT.

OUR RESPONSE: THE TITLE FOR "Abstract" IS NOT ALL CAPITALIZED IN THE REVISED MANUSCRIPT (#3 the most recent version in the system). OTHER SECTIONS TITLES DO NOT APPLY.